Author's response to reviews

Title: Integrated Syphilis/HIV Screening in China: A Qualitative Analysis

Authors:

Joseph D Tucker (jtucker4@partners.org)
Li-Gang Yang (yanglg3@yahoo.com)
Zheng-jun Zhu (smile1960@21cn.com)
Bin Yang (yangbin101@hotmail.com)
Yue-pin Yin (yinyp@ncstdlc.org)
Myron S Cohen (mscohen@med.unc.edu)
Xiang-Sheng Chen (chenxs@ncstdlc.org)

Version: 2 Date: 18 September 2009

Author's response to reviews:

Dear BMC Health Services Research Editors:

Thank you for considering the manuscript entitled, “Integrated Syphilis/HIV Screening in China: A Qualitative Analysis.” As per the recommendation of Dr. Titmus, we have revised the manuscript to make the methods and results sections more clear and consistent with the RATS guidelines.

The revised manuscript is attached.

Sincerely,

Joseph Tucker